Author's response to reviews

Title: Predictors of mortality of patients newly diagnosed with clinical type 2 diabetes: a 5-year follow up study

Authors:

Niels de Fine Olivarius (olivarius@sund.ku.dk)
Volkert Siersma (siersma@sund.ku.dk)
Anni BS Nielsen (annibrit@sund.ku.dk)
Lars J Hansen (larshan@sund.ku.dk)
Lotte Rosenvinge (lotte.rosenvinge@dadmnet.dk)
Carl Erik Mogensen (carl.erik.mogensen@ki.au.dk)

Version: 7 Date: 10 August 2010

Author’s response to reviews:

In this very last proof I have only made a few changes in the titles of the Tables. These titles have been changed by you after my last submission yesterday, and I believe the changes are disturbing the meaning of the tables:

- The word "Cox models" have been deleted from the footnotes of Table 1 and 3.
- The title of table 4 have been changed, so that it consists of only one sentence.

No other changes have been made, and I confirm that the MS is acceptable as it is now.

Sincerely yours

Niels Olivarius